Author's response to reviews

Title: Women-focused development intervention reduces delays in accessing emergency obstetric care in urban slums in Bangladesh: a cross-sectional study

Authors:

Shamsun Nahar (shamsun203@gmail.com)
Morshed Banu (morshed.b@brac.net)
Hashima-E Nasreen (nasreen.h@brac.net)

Version: 5 Date: 27 January 2011

Author's response to reviews: see over
Date 27 January, 2011

To
Melissa Norton, MD
Editor-in-Chief
Journal of BMC series or Pregnancy and Childbirth
BioMed Central Ltd.
236 Gray’s Inn Road, London
WC1X 8HL, UK

MS: 4811879423918786, Women-focused development intervention reduces delays in accessing emergency obstetric care in urban slums in Bangladesh: a cross-sectional study”.

Shamsun Nahar¹, MorshedBanu², Hashima-E-Nasreen²

We would like to take the opportunity to thank you for taking a favorable decision for publication of our manuscript “Women-focused development intervention reduces delays in accessing emergency obstetric care in urban slums in Bangladesh: a cross-sectional study” (MS: 4811879423918786). To speed up publication we thoroughly reviewed, revised and made change on final version of our manuscript. These are detailed, point by point at the end of this cover letter below.

We now submit the final version manuscript for your deliberation and hope that it will meet all requirements for publication. We take this opportunity to thank you again for your time and interest.

Yours sincerely
MorshedBanu
Research Associate
BRAC Research and Evaluation Division
(http://www.brac.net/research)

Editorial comments: Thank you very much for you valuable comments on this manuscript.

1. List of authors in manuscript has been written exactly as in the submission system, the name is written ‘Hashima E Nasreem’ instead of Hashima-E-Nasreen.

2. All references cited in the text are in journal style and correction has been made on the page 5, 8, 11 and 19 and table 1. Full citation details for reference 1, 16 and 28 are given in the reference list.

4. We removed all highlighting in the text and tables.

5. Abbreviations has been formatted in to a paragraph and inserted before the competing interest as recommended.

6. Figure has been cropped and tried our best to minimize white space around the image and visible vertical line has been removed from tables.